Author's response to reviews

Title: Cardiogenic shock as a complication of acute mitral valve regurgitation following posteromedial papillary muscle infarction in the absence of coronary artery disease.

Authors:

federico bizzarri (federico.bizzarri@uniroma1.it)
consalvo mattia (consalvo.mattia@uniroma1.it)
massimo ricci (punkindoctord@libero.ito.it)
flaminia coluzzi (fsalminia.coluzzi@uniroma1.it)
vincenzo petrozza (vincenzo.petrozza@uniroma1.it)
giacomo frati (fraticello@inwind.it)

Version: 8 Date: 28 October 2008

Author's response to reviews: see over
To the Editor,

Dear Editor,
We would like to resubmit for publication the revised paper entitled "Cardiogenic shock as a complication of acute mitral valve regurgitation following posteromedial papillary muscle infarction in the absence of coronary artery disease".
We added another author: Giuseppe Pugliese. I put all the working references and addresses in English.
Hoping that this is the final version of our paper for publication we thank you in advance for the time spent in revising the manuscript…..!!!!!!

Federico Bizzarri